Author's response to reviews

Title: Determining the association of medical co-morbidity with subjective and objective cognitive performance in an inner city memory disorders clinic: a retrospective chart review

Authors:

Corinne E Fischer (FISCHERC@smh.ca)
Depeng Jiang (jiang@cc.umnitoba.ca)
Tom A Schweizer (SchweizerT@smh.ca)

Version: 5 Date: 15 December 2010

Author's response to reviews: see over
Dear members of the editorial production team:

Re: Manuscript 9213588704094178

Determining the impact of medical co-morbidity on subjective and objective cognitive performance in an inner city memory disorders clinic: a retrospective review

Thanks very much to the editorial production team for your editorial comments. I will attempt to respond to each point in turn.

With respect to the comments from the editorial production team:

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

In response to the editorial production team’s comment all of the author’s affiliations have now been included on the title page of the manuscript, page 1.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, ?The Title of My Paper: A Scientific Study.? should be changed to ?The title of my paper: a scientific study?

The change requested by the editorial production team has been made to the manuscript as noted on page 1, the title page.

Tracking: please remove all tracking from the manuscript

As requested by the editorial production team all tracking has now been removed from the manuscript.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

As requested by the editorial production team all typographical errors have now been corrected.
Thanks very much to the editorial production team for your helpful comments. Please let us know if anything requires further clarification.

Sincerely,

Corinne Fischer MD FRCPC
Director of the St. Michael’s Hospital Memory Clinic